Author's response to reviews

Title: Neuroprotective efficacy and therapeutic window of curcuma oil: in rat embolic stroke model

Authors:

Preeti Dohare (preetidohare13@gmail.com)
Puja Garg (pujagarg1jsr@rediffmail.com)
Uma Sharma (uma_1969@hotmail.com)
N. R. Jagannathan (jagana_1954@hotmail.com)
Madhur Ray (madhurray.cdri@gmail.com)

Version: 4 Date: 13 September 2008

Author's response to reviews:

MS: 2097858650178912

Neuroprotective Efficacy and Therapeutic Window of Curcuma oil: In Rat Embolic Stroke Model; Preeti Dohare, Puja Garg, Uma sharma, N. R. Jagannathan and Madhur Ray

To

The BioMed Central Editorial Production Team

We have tried to meet the requirements spelled out in your letter to the best of our efforts. We sincerely hope we have been able to do the needful.

Sincerely yours,

M Ray

Formatting changes requested

1. Please can we -------------------------- competing interests.'
Answer: changes done.

2. Title - Please use -------------------------- study
Answer: needful is done

3. Affiliations -------- include full stops in abbreviated country names.
Answer: needful is done.

4. Keywords and Running head ¿ Please remove these.
Answer: needful done.
5. Abstract ------------------what is in the submission system.
Answer: needful done.

6. Figure title - ------------------------------------------------------- uploaded.
Answer: needful is done.